Author's response to reviews

Title: Qualitative evaluation of primary care providers experiences of a training programme to offer brief behaviour change counselling on risk factors for non-communicable diseases in South Africa

Authors:

Zelra Malan (zmalan@sun.ac.za)
Robert Mash (rm@sun.ac.za)
Katherine Everett-Murphy (km.murphy@uct.ac.za)

Version: 8 Date: 5 August 2015

Author's response to reviews:

Comment
We notice that table 3 contains personal information on the study participants. This information cannot be published unless consent to publish was previously obtained from all participants. If this is the case, please state this in your manuscript. If not, please remove this table from the manuscript.

Response
Thank you, in line 244 the sentence has been added and highlighted